Author's response to reviews

Title: Tuberculous peritonitis in German Patient with PBC

Authors:

Yilin Vogel (yilin.vogel@marienhospital-herne.de)
Jan C Bous (jan.bous@marienhospital-herne.de)
Guido Winnekendonk (guido.winnekendonk@marienhospital-herne.de)
Bernhard F Henning (bernhard.henning@rub.de)

Version: 4 Date: 13 November 2007

Author's response to reviews:

Dear Sirs,

Following a thorough revision of our manuscript "Tuberculous peritonitis in a German patient with primary biliary cirrhosis (PBC): a case report" in accordance with examiner 1 and examiner 2 we are now sending you the corrections.

We have in particular reduced the number of illustrations according to the recommendation and subjected the text to a comprehensive linguistic revision. The discussion now takes those points mentioned by examiner 2 into account.

We would be delighted to receive a positive response from you and remain

Yours sincerely,